Author's response to reviews

Title: Immunologically reactive M. leprae antigens with relevance to diagnosis and vaccine development.

Authors:

Lucas H. Sampaio (lucashfs@gmail.com)
Mariane M. A. Stefani (mstefani@iptsp.ufg.br)
Regiane M. Oliveira (remorillas@gmail.com)
Ana L. M. Sousa (anamaroclo@gmail.com)
Greg C. Ireton (gireton@idri.org)
Steven G. Reed (sreed@idri.org)
Malcolm S. Duthie (mduthie@idri.org)

Version: 7 Date: 17 January 2011

Author's response to reviews:

Minor changes have been made in the manuscript and also in Fig 5.